Author's response to reviews

Title: A novel SNP analysis method to detect copy number alterations with an unbiased reference signal directly from tumor samples

Authors:

Alex Lisovich (all67@pitt.edu)
Uma R. Chandran (chandranur@upmc.edu)
Maureen A. Lyons-Weiler (lyonsweilerma@upmc.edu)
William A. LaFramboise (laframboisewa@upmc.edu)
Ashley R. Brown (arg39@pitt.edu)
Regina I. Jakacki (regina.jakacki@chp.edu)
Ian F. Pollack (ian.pollack@chp.edu)
Robert W. Sobol (rws9@pitt.edu)

Version: 3 Date: 19 January 2011

Author’s response to reviews: see over
Wednesday, January 19, 2011

Dr. Miguel Angel Pujana, Associate Editor &
Prof. Debashis Ghosh, Section Editor
BMC Medical Genomics

Dear Drs. Pujana and Ghosh:

Please find enclosed our final version, re-revised manuscript (MS: 2639767364203720) entitled, “A novel SNP analysis method to detect copy number alterations with an unbiased reference signal directly from tumor samples” for publication in BMC Medical Genomics by Alex Lisovich, Uma R. Chandran, Maureen A Lyons-Weiler, William A. LaFramboise, Ashley R. Brown, Regina I. Jakacki, Ian F. Pollack and Robert W. Sobol

In response to your e-mail of 01-19-11, we have inserted into the manuscript the accession number for our publically available data. In addition, we have gone through the manuscript-formatting checklist one more time to ensure that our revised manuscript conforms to all of the points. We look forward to hearing from you with regard to publication.

Sincerely,

Robert W. Sobol, Jr., PhD
Assistant Professor
Departments of Pharmacology & Chemical Biology and Human Genetics
Co-Director, UPCI Vector Core Facility
Director, UPCI Lentiviral Facility
University of Pittsburgh Cancer Institute